Author's response to reviews

Title: p-Glycoprotein ABCB5 and YB-1 expression plays a role in increased heterogeneity of breast cancer cells: correlations with cell fusion and doxorubicin resistance

Authors:

Ji Yeon Yang (warum33@catholic.ac.kr)
Seon-Ah Ha (hsa387@hanmail.net)
Yun-Sik Yang (ysyang@wonkwang.ac.kr)
Jin Woo Kim (jinwoo@catholic.ac.kr)

Version: 10 Date: 22 July 2010

Author's response to reviews: see over
Research Article: \( p \)-Glycoprotein ABCB5 and YB-1 expression plays a role in increased heterogeneity of breast cancer cells: correlations with cell fusion and doxorubicin resistance

Dear The BioMed Central Editorial Team:

I thank the editors of the BMC Cancer for taking their time to review our article. We have made corrections according to the editorial adviser’s comments. The responses to adviser’s comments are detailed below and the changes were made in the manuscript.

Yours Sincerely,

Prof. Jin Woo Kim, M.D. & Ph.D.
Catholic Research Institute of Medical Science, College of Medicine, The Catholic University of Korea, 505 Banpo-dong, Seocho-ku, Seoul 137-040, Korea. Tel: +82-2-590-2389; Fax: +82-2-593-2389; E-mail: jinwoo@catholic.ac.kr

Reviewer's comments:

1. Figures: Figure legend 1 contains graphics. These will either need to be included within the image as a key, or written in text (e.g. "black square").
2. Figures: The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order.

….Each MCF-7 subtype, labeled with either EGFP/YB-1 or DsRed/YB-1, were co-cultured as a single monolayer (Figure 4A). Figures 3A, B and figure 4B show that some of these cells were fused……

Changed to Each MCF-7 subtype, labeled with either EGFP/YB-1 or DsRed/YB-1, were co-cultured as a single monolayer. Figures 3A, B, figure 4A and B show that some of these cells were fused…. 

3. All figures must be cited; figures not cited will be excluded from the final manuscript.  
Changed to Figure 3C and 5B are cited 

4. Figure 3 is currently cited before figure 2 in the manuscript.  
I can not find out “Figure 3 is currently cited before figure 2 in the manuscript“  

5. Competing interests: Please format the 'Competing interests' to include all authors. If there are none to declare, please write 'The authors declare that they have no competing interests'.

Changed to The authors declare that they have no competing interests